Author's response to reviews

Title: In vitro digestion and lactase treatment influence uptake of quercetin and quercetin glucoside by the Caco-2 cell monolayer

Authors:

Jeanelle Boyer (rl23@cornell.edu)
Dan Brown (RL23@cornell.edu)
Rui Hai Liu (RL23@cornell.edu)

Version: 2 Date: 31 December 2004

Author's response to reviews:

1. Affiliations: Corrected as suggested. Thank you.
2. Line numbers were removed as suggested.
3. Figures: Changes made as suggested.